**Original Article** 



A virtuous circle: Stakeholder perspectives of a short-term intensive parent training programme delivered within the context of routine services for autism in China Autism 2022, Vol. 26(8) 1973–1986 © The Author(s) 2022

Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/13623613211070869 journals.sagepub.com/home/aut

# Zuyi Fang<sup>1</sup>, Jamie M Lachman<sup>2,3</sup>, Cheng Zhang<sup>4</sup>, Dongping Qiao<sup>1</sup> and Jane Barlow<sup>2</sup>

#### Abstract

Although the evidence of parent training programmes for families of autistic children has continued to grow, little is known about the experiences and perceptions of key stakeholders, especially in low- and middle-income countries. This qualitative study was part of a larger real-world evaluation of a short-term intensive parent training programme in routine services delivered to caregivers of autistic children aged 3–6 years in China. It aimed to provide insights into programme acceptability, reasons for participant involvement, and factors related to programme implementation. Fourteen caregivers participated in the in-depth interviews, and two focus group discussions were conducted with eight practitioners. Data were analysed using a combination of data- and theory-driven approaches. Findings point to the needs in low autism resource settings for substantial practice and feedback; group support; individualised coaching; more autism-related knowledge, resources and activities for children and extended family members; and organisational support to practitioners. Further research is suggested to address the recommendations and assess their effectiveness empirically.

#### Lay abstract

While much knowledge about autism derives from high-income countries, most people diagnosed with autism reside in low- and middle-income countries, where little is documented in terms of local interventions. This is also true for parent training programmes for families of autistic children. An evaluation was conducted to understand the effects of a short-term intensive parent training programme delivered in routine services for families of autistic children in China. This study reported results from the in-depth interviews with 14 participating caregivers and group discussions with eight group leaders. The interviews and discussions were aimed at learning (1) to what extent the programme components were deemed acceptable, (2) what affected caregivers' attendance and engagement in the programme and (3) what affected group leaders' delivery of the programme. Findings suggested that future parent training programmes provide adequate opportunities for caregivers to practice and receive feedback; group support; coaching experience tailored to individual challenges; more autism-related knowledge, resources and activities for children and extended family members; and organisational support to group leaders. This study highlights the value of qualitative research and points to the need for more empirical studies to address the recommendations, so that research findings can be better utilised to promote practices.

#### **Keywords**

autism spectrum disorders, family functioning and support, interventions – psychosocial/behavioural, pre-school children, qualitative research

<sup>1</sup>Beijing Normal University, China

<sup>2</sup>University of Oxford, UK

<sup>3</sup>University of Glasgow, UK

<sup>4</sup>University of Leeds, UK

#### **Corresponding author:**

Zuyi Fang, School of Social Development and Public Policy, Beijing Normal University, 19 Xinjiekou Wai Street, Beijing, 100875, China. Email: fangzuyi@outlook.com A large regional study recently conducted in China showed a prevalence rate for autism of 1% (approximately 3.1 million children), which is in line with the estimate for Western countries (National Bureau of Statistics of China, 2021; Sun et al., 2019). Families of autistic children in low- and middle-income countries (LMICs) tend to face more accumulated adversities than those in high-income countries (HICs), including problems such as resource constraints and social marginalisation (Walker et al., 2011). Autistic children are more likely than their non-autistic cohorts or cohorts diagnosed with other developmental disabilities to experience co-occurring externalising behaviours (Totsika et al., 2011). These behavioural challenges are linked to increased parenting stress and have an impact in terms of child development (Hayes & Watson, 2013).

Early interventions have been advocated to optimise child outcomes in autism (Courchesne et al., 2011). Early behavioural treatment models for young autistic children, such as Early Intensive Behavioural Interventions and Early Start Denver Model, have shown promising effects in terms of improving behaviours, cognitive performance and language skills (Warren et al., 2011). However, the intensive delivery over several consecutive years makes such programmes unfeasible for many families. Moreover, the common shortage of specialist providers and infrastructure in low-resource settings limits access to intensive specialist care (Morris et al., 2011). As such, relatively shorter parent training and education has been identified to be an important strategy to address this significant unmet need (Zwaigenbaum et al., 2015).

Components of parent training programmes differ, ranging from information sharing to teaching primary caregivers parenting techniques and involving them as the change agent for child outcomes. Systematic reviews have been conducted to evaluate the effectiveness of such programmes in terms of promoting child behavioural adjustment, improving parent-child interactions, reducing autism symptoms and promoting child skills (Fang et al., 2020; Nevill et al., 2018; Oono et al., 2013; Parsons et al., 2017; Postorino et al., 2017; Ratliff-Black & Therrien, 2020; Strauss et al., 2013). Fang et al. (2020) also focussed on programmes in China, found no significant group differences between programmes that originated outside and within China or between programmes with and without clinically qualified deliverers in terms of reducing child challenging behaviour, thereby supporting the use of parent training across cultures and resource settings.

While the growing evidence base for such programmes is promising, much less is known about the views of key stakeholders with regard to the acceptability and feasibility. Qualitative research can identify components that are meaningful to beneficiaries and that have the potential to maintain participant engagement (Butler et al., 2020). However, there are few qualitative studies focussed on autism, especially in LMICs (Makombe et al., 2019). This current study aims to address this gap by investigating the experiences and perceptions of key stakeholders with regard to a shortterm intensive parent training programme, named SREIA (Stars and Rain Education Institute for Autism), which has supported thousands of families of young autistic children in China over the course of two decades.

# **Research question**

This qualitative study was aimed at answering the following overarching research question: What are stakeholders' experiences of participating in and delivering the SREIA programme within a routine service context? It was investigated with specific reference to the extent to which the programme content and delivery methods were deemed to be acceptable; and the extent and nature of the facilitators of and barriers to participant involvement and programme implementation (Berkel et al., 2011).

# **Methods**

## Study design

This qualitative study was embedded within a larger quasi-experimental evaluation of the SREIA parent training programme, delivered from September to November 2020 (ClinicalTrials.gov CT04257331). The programme was shown to be associated with improvements in child behavioural adjustment, parental mental health, parental knowledge and parental over-reactivity (under review). In-depth interviews and focus group discussions (FGDs) were conducted immediately after the completion of the programme to provide insights into the intervention process and outcomes (Creswell et al., 2011). Ethical approval was provided by the University of Oxford (R67619/RE001) and Beijing Normal University (SSDPP-HSC2020001).

#### Programme

The SREIA short-term intensive parent training programme has been delivered to Chinese families of autistic children aged 3-6 years since 1993 by a long-established non-governmental organisation (NGO). It is based on applied behaviour analysis, social learning theory, operant conditioning, developmental and cognitive behaviour theory. It is aimed at reducing child challenges and promoting child development by increasing parental knowledge, improving parenting skills, changing parental attitudes, promoting parental mental health and creating social support. A range of techniques are delivered using a variety of methods in an intensive format, involving 5 days a week for 11 consecutive weeks (Table 1). Such an intensive approach is feasible in China due to the fact that around two-thirds of Chinese families of autistic children have at least one parent who is a full-time caretaker (Guo et al., 2014). Chinese parents are often the primary instructor and

Table I		SREIA	programme	topics	and	delivery	/ technia	ues.
i abic i	•		pi ogi annine	copies	and	denivery	ceeiiiiq	acs.

Topics	Delivery techniques
Knowledge of ASD and child development	Individual consultation
	Didactic presentation
Child behavioural adjustment and behaviour management techniques based on ABA, including reinforcement (praise and rewards), prompts, fading, shaping and chaining	Practitioner observation
TEACCH – 'Structured TEACCHing'	Facilitator modelling new techniques
Instruction giving	Cases and stories
Ignore	Practice in sessions and in vivo feedback
Redirect	Role play
Parent-child communication and interaction (including parent-child play)	Homework and feedback
Emotional labelling	Videorecording and feedback
Fine and gross motor skills	Group discussion on skills and experiences
Communicate with family members	Group problem-solving
Parental self-regulation	Physical activity
Parental attitude and expectation	Reinforcement and active listening by practitioners
Skill maintenance and generalisation	Online group communication
Plan for the future	Informal peer support group
	Printed materials
	Childcare

ASD = autism spectrum disorder; ABA = applied behaviour analysis.

advocator for their autistic child, which highlights the need for such programmes to increase their self-efficacy.

Individual pre-consultations are offered with the aim of enabling programme practitioners to better understand family strengths and concerns. An individualised treatment plan is created for each parent-child dyad, covering targeted child behaviours, caregiver areas for improvement and other priorities identified by caregivers. Fourteen 2-h didactic presentations on the core themes are offered across the 11 weeks, with one or two topics each week. The large didactic presentations are then followed by additional training and in vivo practice in small groups of ten dyads to enhance the understanding and use of skills. Content delivered in small caregiver groups focus on the weekly topics, but are also tailored to various degrees for each group to address different needs. Group sessions often start with the practitioner revisiting the content in that week's didactic presentation(s) and modelling the new techniques, followed by caregivers practicing the techniques in vivo and receiving comments from practitioners and other caregivers. Caregivers are assigned homework activities to practice after sessions. They are encouraged to submit written or video records of their practice at home for practitioners to provide feedback. Communications between caregivers and practitioners remain open and flexible. Opportunities for peer interactions are created by arranging group discussions, group problem-solving activities and an online social and messaging group. The programme is primarily delivered in person, with the 14 didactic presentations also streamed online to avoid mass gathering during COVID-19. The pandemic also led to a lack of childcare services, which used to be provided during the large didactic presentations. The programme had two versions of delivery, which were implemented concurrently and share the same programme content, with the purpose of exploring ways to reduce cost while improving outcomes. Parent groups in Version A attend didactic sessions together in a larger group but practice sessions separately, whereas in Version B, parents have all sessions delivered separately for each group.

#### Participants

Participants were drawn from practitioners and families involved in the SREIA programme delivered between September 2020 and November 2020. All programme practitioners (N=8) were invited and consented to participate in the FGDs. They were employees of the SREIA implementing organisation and included Han females in their 20s–40s. They were from diverse educational backgrounds, having college or associate degrees unrelated to healthcare or autism, such as in social work and early education. Five of the eight practitioners had more than 5 years of experience in delivering the programme, whereas the other three practitioners were relatively young. All had been trained and certified by the implementation agency to deliver the programme. The training for the old and new practitioners differed, as the training has evolved over time, although the primary focus remained on knowledge about autism, programme content and programme delivery.

Participant characteristics	Sample N = 14				
Caregiver age	26-50 years, M=34 (5.94)				
Caregiver education	University or higher	28.6% (4)			
	Associate degree	42.9% (6)			
	High school	14.3% (2)			
	Prefer not to tell	14.3% (2)			
Relationship with	Mother	92.9% (13)			
the child	Grandmother	7.1% (1)			
Ethnicity	Han	92.9% (13)			
	Man	7.1% (1)			
Work status	Working	28.6% (4)			
Spouse work status	Working	92.9% (13)			
Family income	< 60k (Chinese currency)	57.2% (8)			
Permanent residency	Rural	57.1% (8)			
	Urban	42.9% (6)			
Child age	3–6 years, M=4.76 (0.98)				
Child gender	Male	78.6% (11)			
Previous services	Day care	21.4% (3)			
	Individual therapy	78.6% (11)			
	No service attended	7.1% (1)			

Table 2. Caregiver participant characteristics.

The fidelity of their implementation was assessed using practitioner checklists. Practitioners reported whether they delivered the core components prescribed in the schedule each week. The fidelity scores (maximum=1) comprised the overall ratio of delivered components to prescribed components. Quality of delivery was also measured by research staff observing parent group sessions and completing a quality of delivery survey. The survey contained 24 items, assessing the general skills of facilitating parent groups (such as knowledge of content, teaching skills and sensitivity to participants' feelings and experiences) and skills of modelling parenting techniques and facilitating practice sessions (such as appropriately setting up the space and providing adequate support during participant practice). A total score is summed (maximum=72). Two sessions for each practitioner were randomly selected and recorded for research staff to rate the quality of delivery. Their overall fidelity rate was 0.95 and the delivery was of medium quality, with a mean score of 51.86 (SD=1.13).

The programme was delivered to seven groups of 10 caregiver-child dyads – four groups in Version A and three groups in Version B. All participating children were diagnosed with autism aged between 3 and 6 years. Evidence shows that caregiver involvement is linked to the potential of programmes to produce desirable outcomes in interventions for autistic and non-autistic children (Baydar et al., 2003; Carr et al., 2016; Nix et al., 2009). As such, participants for the in-depth individual interviews were identified based on their overall attendance and engagement as rated by the programme practitioners. The practitioners recorded caregiver attendance using an attendance register and judged their engagement on a daily basis according to four

criteria: responsiveness to practitioners, interactions with other caregivers, participation in group discussions and completion of homework activities. Three caregivers with the highest and lowest levels of involvement were identified in each parent group. However, it should be noted that the rates of attendance (0.98 out of 1) and engagement (5.58 out of 6) were generally high. Recruitment was conducted with the help of practitioners and stopped when data saturation – no additional data are found that can contribute to the emergence of new codes or themes (Glaser & Strauss, 1967) – was reached. The final sample consisted of 14 caregivers (eight from Version A and six from Version B) with different levels of involvement. Table 2 presents the characteristics of the caregivers. Oral informed consent was obtained from all participants.

#### Data collection and analysis

The interviews and FGDs were conducted during the week immediately after the completion of the intervention. The caregivers were interviewed individually for a maximum of 1.5 h. Two FGDs were conducted, lasting for approximately 2 h (N=4 practitioners per group). Semi-structured interview schedules were designed around acceptability, involvement and implementation. Each topic was started with broad open-ended questions followed by probes to evoke detailed responses. Questions evolved over the course of the study to adapt the interview approach based on participants' responses (Green & Thorogood, 2014). The FGDs were facilitated by the first author and the caregiver interviews were conducted by the first author and a trained interviewer with a master's degree in social work.

The analytic approach was underpinned by a contextualist framework, which involved combining the traditionally data-driven approach with a theory-driven approach (Braun & Clarke, 2006). As such, coding of the data was guided by a coding framework, while also allowed themes to emerge from the data, capturing the full range of responses. This hybrid approach has demonstrated rigour in previous research (Fereday & Muir-Cochrane, 2006). The data analysis integrated the development of a coding framework into the six phases of thematic analysis described by Braun and Clarke (2006). It involved the following six steps. First, several coding domains were outlined based on the theoretical concepts (programme acceptability, participant involvement and programme implementation) and research questions (facilitators and barriers; Boyatzis, 1998). In addition to these coding domains, a topic pertaining to the experience of having an autistic child in China emerged during data collection. Second, the transcripts were read and re-read to create a list of coding ideas. Third, initial codes were generated from a subset of data and organised in terms of the coding domains. Fourth, preliminary themes were created to develop an initial coding framework. Fifth, the coding framework was applied and constantly modified throughout the coding process. Finally, themes were refined and reported.

#### Reflexivity, quality and trustworthiness

The first author introduced the study to the participants and coordinated interview times. The interviews were conducted by the first author and a trained interviewer, who are both females, born and raised in China. The first author is a PhD candidate at an academic institute in the United Kingdom, and has been researching parent training programmes for children with and without disabilities. The other interviewer completed a master's degree in social work in China, and has experience in working with families of autistic children and conducting qualitative interviews.

Given the diverse backgrounds of interview participants, we promoted reflexivity of this study by recognising and addressing power differentials between interviewers and participants. All interviews were conducted in a friendly and empathetic manner. Rapport was built prior to asking any questions. The interviews started with casual conversations, informing participants what to expect in the interview and addressing their concerns about confidentiality. The interviews were conducted in a space that was familiar to the participants. The purpose and anonymisation of recording, as well as their rights to erase the recording and/or withdraw the data, was fully explained. Light refreshments were also provided to create a welcoming and relaxed environment. Further, the coding software ATLAS.ti was used to document the entire coding procedures, including any evolving perceptions and critical decision points. The involvement of second coder and the discussion of discrepancies also facilitated the introspection process.

In addition, several steps were taken to improve the quality and trustworthiness of the results including the data being collected from both caregivers and facilitators and transcribed data being sent to participants for verification and additional inputs before data analysis was begun. To further improve credibility, data analysis was conducted by another researcher checking 50% of the initial coding categories, participating in the development of the coding framework and reviewing all codes and themes before they were finalised.

#### Community involvement

The implementation organisation took part in the planning of the overall research project, by contributing to the development of the research design, research questions and interview tools. The recruitment of participants was undertaken by the organisation using its existing resources. The programme was delivered by staff of the organisation and took place at the SREIA venues. The organisation did not take part in the data collection or analysis. However, it has been actively involved in the interpretation of results and taking the lead in conveying the research findings to a wider autism community in China. No other organisations were involved in this study.

## Results

A summary of the themes and subthemes under each domain is presented in Figure 1 and discussed in the following sections. The identifiers of caregiver quotes start with C and those of practitioners start with FGD. Additional quotes for each theme are presented in the Online Appendix.

# Perceived changes

The interviewees described numerous ways in which the programme had produced changes in themselves, their child, parent–child relationships and other family members.

Caregivers were viewed as having *improved mental wellbeing* after participating in the programme, as a result of having better emotion regulation, moral support from peers and practitioners and a sense of hope in relation to the progress in their child. Caregivers also reported increased acceptance of their child's diagnosis:

I haven't even told my parents. I couldn't accept that my son has Autism. I was afraid that we would be discriminated. Now I feel much more peaceful. Last week my neighbour asked me where we went, and I told her . . . I plan to tell my parents. (CH3, 5-year-old boy)

Caregivers described *increased parental self-efficacy* characterised by more positive parenting practices and less harsh discipline. In addition, they highlighted the ability to make decisions about which other services to use as a result of having more knowledge about autism.

Perceived changes in children included better emotional and behavioural adjustment and more adaptive living skills, which were perceived to be the result of more positive parenting practices and an increase in child communication skills. More parent-child interactions and affection were commonly referred to by the caregivers, which were similarly considered as being attributable to positive parenting and an increase in parent-child play.

Some caregivers who shared information from the programme with their partner described *positive changes in partners*, such as less stress and more collaborative parenting. These perceived positive changes may have created a 'virtuous circle' in which the occurrence of one led to that of another, which further promoted a continuous process of improvement. These interactions were also well described by one of the practitioners:

What we are hoping to do is to create a virtuous circle . . . The core is centred around getting to know your child, know his deficit, and the signals he tries to give you through his



Figure I. Overview of themes.

behaviours . . . based on the signals, you apply the skills. During the process, you will know even more about your child, and see all the challenges he has been struggling with, then you know that you should control your own emotions and help him . . . You will have more strength and confidence, which in turn improves your emotions . . . With better emotions, you will also communicate with your family in a better way . . . then there is a better environment at home for your child to develop . . . So you can see there is a virtuous circle – improvement in one area can ultimately lead to the improvement of their overall functioning. (FGD2)

Despite attempts to share their learning, there was generally perceived to have been a *lack of change in elderly family members*:

I've tried to explain, Autism and the techniques, to his grandparents, but I don't see any changes . . . Some concepts, like Autism, are just difficult for them to understand. In their opinion, if a child does not communicate, it's only because he doesn't want to. No other excuse. (CH8, 5-year-old boy)

#### Programme acceptability

**Programme content.** All aspects of the programme were perceived to be helpful by both parents and practitioners, with many of the techniques being seen as interrelated. The session that focussed on *knowledge about autism* at the beginning of the programme was perceived to be central to understanding the diagnosis and its implications, thereby promoting the acceptability of techniques taught at later stages:

[the knowledge of Autism] is like the central nervous system, without which you won't be able to receive other information or coordinate any activity. (FGD1)

Those who skipped the session always don't accept their children and are more sceptical about whatever we say. (FGD2)

Skills reported by the caregivers as being the most likely to be used included *behaviour management techniques, visual supports* and *daily structure and routine*, which were consistent with the perceived changes in children, such as fewer internalising and externalising behaviours and improved communicational skills.

#### Delivery methods

A number of delivery methods were perceived to have been important in establishing new skills in caregivers, including *modelling, practice* and *feedback*, as well as a combination of *group experience* and *individualised support* to promote social inclusion and address diverse concerns.

The caregivers and practitioners both highlighted the importance of practitioner modelling, which was perceived to have involved a demonstration of the application of skills, build trust and promote caregiver understanding of their child:

... it made me see how it was like for my child to interact with other people. (CL5, 3-year-old boy).

... because they would only believe us when they saw us successfully applying the skills to their child. (FGD2)

In-session practice and in vivo feedback were also identified as being central to their learning, as they were perceived to have scaffolded caregivers' engagement with targeted skills. The caregivers also explained that the parent groups provided an experience of social inclusion and belongingness, which was felt to have nurtured a greater acceptance of their children and themselves:

(the caregiver group) were like a family . . . We learn from each other about how we should treat our child and ourselves. (CH1, 5-year-old girl)

Ongoing and flexible communication was also referenced by both groups of participants. The constant close communication between practitioners and caregivers was described as being like one-to-one coaching:

They coached me on things that my child needed most. Every time I thought I could not make it, I messaged her, and she would rebuild my confidence. (CH6, 4-year-old boy)

Furthermore, the flexibility within the programme was described by the practitioners as being core to implementation success, in that the programme provided sufficient space to accommodate individual need. Other aspects of the delivery that were described as being helpful by the caregivers included *dedicated practitioners* and *activities for children*. The practitioners were perceived to be knowledgeable, responsive to caregivers' concerns and as treating participants equally. *Activities for children* freed caregivers from childcare:

That is when I felt most relaxed. I take care of him 24/7. Only when they were doing activities, I could catch a breath. (CH7, 3-year-old boy)

# Facilitators and barriers to participant involvement

*Facilitators*. Facilitators of participant involvement included *social support* and *intrinsic motivation*. Caregivers emphasised social support as a major facilitator of their involvement, especially with regard to programme enrolment. Support from a variety of sources – spouses, extended families, relatives, peers and employers – made enrolment possible. Social support was also perceived to have promoted participant engagement by enhancing their learning experience through collective problem-solving. They also considered familial support as being necessary to help them to

maintain consistency of practice at home. In terms of intrinsic motivations, caregivers overwhelmingly talked about a sense of urgency, as they were aware of the importance of early intervention and the role of caregivers in influencing their child's development. Children's progress was commonly identified as another major reason for continuing to attend the sessions, as it was described as motivating their learning and use of skills.

Barriers. Participant involvement was perceived to have been inhibited by (1) obstacles within the family, (2) problems with the programme design and logistics and (3) caregiver negative beliefs and emotions.

Lack of family support was one of the barriers to involvement described by caregivers who described that grandparents were not exposed to any programme content and did not understand the diagnosis or new skills, thus limiting wider support from within the family.

In terms of the programme design and logistics, caregivers felt that it was not long enough to support their learning of targeted skills, pointing to the need for more opportunities to practice and receive feedback, and also for the opportunity to address their lack of confidence:

Less than three months . . . We had no chance to be supervised on many of those skills . . . We will need to keep practicing and refer to the notes. (CH9, 5-year-old boy)

I've heard it and understood it, but I still don't think I can do it well. (CH2, 5-year-old boy)

Children being present in sessions where they were not directly involved in the demonstration and practice of skills was also reported by caregivers to be distracting:

I was about to take some notes and he started to yell. (CH7, 3-year-old boy)

Online delivery during COVID-19 was identified as another barrier by caregivers who were less able to maintain their focus online and whose learning was often disrupted by technological failure. Another perceived common barrier to participant engagement that was caregiver negative emotions, which were often perceived to have been caused by child non-compliance and the lack of family support:

Overall, it's mainly their mental state that determines if they can develop solid skills. . . . Although we practitioners need to guide them towards better mental state, their family environment plays a more decisive part in it. (FGD2)

Pre-existing knowledge of the diagnosis or effective practices also made some caregivers resistant to the introduction of strategies:

(previous service providers) said making the child speak was the most important thing . . . So I skipped some sessions that I thought not related to language skills. I did not realise until now that there can be so many ways for us to communicate. (CL1, 4-year-old boy)

#### Programme implementation

*Facilitators.* Programme delivery was described by practitioners to be facilitated by (1) *caregiver–practitioner cooperation*, (2) *caregiver peer support* and (3) *practitioner peer support*.

Caregiver-practitioner cooperation was perceived by practitioners to improve the programme delivery and be promoted by caregiver trust:

It took less time for us to build trust this time . . . Perhaps it's because the COVID quarantine made them realise what they need to learn . . . Shared goals were established right away and I found it easy to communicate with them . . . (FGD2)

Peer support among caregivers was also perceived to be important in terms of implementation because it increased participants' responsiveness during sessions:

There is a feeling of solidarity among them . . . I don't need to do much to coordinate. They influence each other and help each other. (FGD1)

At the practitioner-level, *peer support among practitioners* involving the discussion of difficult cases and sharing of stress-reduction strategies, was perceived to have improved the quality of delivery.

**Barriers.** The major obstacles to programme delivery that were identified by practitioners were (1) *caregivers' lack of acceptance of their children*, (2) *problems with the programme* and (3) *insufficient organisational resources*.

A lack of knowledge with regard to the diagnosis was viewed as resulting in denial of the autism-related deficits, which was in turn perceived to make it difficult to convince caregivers of the relevance of the skills during the delivery of the programme:

Sometimes they don't accept the deficits. Even if the child couldn't speak, she wouldn't use pictures (to communicate), because otherwise she was admitting that her child was different. Other people would also notice his difference . . . It's hard to deliver unless they accept the diagnosis. (FGD2)

In terms of programme design, practitioners highlighted the dilemma of having children present in the sessions. As with the caregiver reports, the practitioners noted the distractions caused by children being present, but also stressed that their presence allowed for substantial opportunities to practice new skills. The practitioners further discussed the difficulties of having to cover a wide range of topics. Thus, even though all the content was perceived as being helpful and necessary, having too many topics within one programme was also felt to be overwhelming and challenging.

At an organisational level, the heavy workload and lack of organisational support were highlighted as significant barriers to programme delivery. The shortage of personnel was perceived to have resulted in an intense workload and high levels of practitioner stress. Practitioners also described an absence of formal supervision and mental health support. This may explain in part why practitioners identified peer support among practitioners as essential for programme delivery.

#### Areas for improvement

**Programme content.** Given that children being present in the session was one of the major barriers to participant involvement, it is possibly not surprising that the caregivers predominantly asked for *additional resources for children*, such as structured activities and childcare, when their child was not involved in modelling and practice. Such activities were also deemed to be an opportunity to develop children's social skills. The caregivers also highlighted the need for *brief training targeting other family members* to reduce resistance with the family and maximise child opportunities to learn:

Other family members also need to learn, but they won't listen to us. They'll listen better if it's someone with more experience. (CH7, 3-year-old boy)

In addition, some participants suggested the need for *more sharing of information* on autism treatments, such as other evidence-based interventions and interventions for older children. They also emphasised the need for *formal mental health support*, such as counselling and referral. Furthermore, they wanted *more instructions about parent-child interaction*:

Parent--child relationship is the foundation . . . We noticed that many caregivers did not know how to play, although they really wanted to interact with the child, they did not know where to start. . .We need to work on that. (FGD2)

Delivery methods. In terms of delivery methods, caregivers identified the need for more *practitioner modelling*, *practice in session* and *in vivo comments* to enhance skills. More *individualised support* was also perceived to be required to address individual concerns.

Caregivers identified a need for the *support to be extended* after the completion of the programme to facilitate the use of skills at home and keep up a positive attitude:

But when the programme ends and we stay back at home, we have to again face all the challenges we had before. It is hard to make sure we still have the strength to do all these (new skills) or even to stay in a good emotional state. (CH5, 4-yearold boy)

Caregivers also highlighted the need for a *better setup* of the learning space, including arranging for an offline venue to better demonstrate skills and improving the quality of online sessions.

# Discussion

As the evidence in relation to autism treatments has evolved, an understanding of the intervention experiences of stakeholders has become more important. This is especially true in LMICs, where most autistic children reside but little is documented in terms of local interventions and how they are provided. To the best of our knowledge, this study is one of the few to investigate the experience of participating in and delivering a short-term intensive parent training programme for families of autistic children in an LMIC. The findings of this study provide valuable information about the acceptability and feasibility of autism treatments and can be used to address the substantial unmet needs of young autistic children for interventions in low autism resource settings where families face similar challenges.

#### Summary and implications

The results indicate that caregivers and practitioners highly valued the comprehensiveness of the programme, describing all topics as being helpful and necessary. This may reflect the general lack of autism information and local services more widely (Clark et al., 2019). Indeed, many parents described the programme as being the major source of information for them and wanted the opportunity to participate in future programmes to secure more resources and support. Two previous studies on parenting interventions for families of autistic children in HICs also showed that parents often valued additional information, such as the technique of social stories and advice on the seeking and selection of services (Dababnah & Parish, 2016; Whittingham et al., 2006). However, practitioners were concerned that the comprehensive cover of the programme could potentially compromise the quality of delivery, suggesting the need to help caregivers maintain focus by encouraging them to closely relate the programme content to their own personal goals.

The results also highlighted the importance of learning about autism, which enabled caregivers to better understand and accept the diagnosis and their children's behaviours, thereby paving the way for learning autism-related parenting techniques. This is consistent with previous research, which showed that introducing autism terms helped establish a 'common ground' (Dababnah & Parish, 2016). Moreover, as parents in China typically receive inadequate guidance on the selection of appropriate treatments for their children (McCabe, 2012), getting to know autism and their child better may support their future decision-making. In accordance with other research that found high caregiver rating of content on behavioural management, visual support and structured routines (Lachman et al., 2016), caregivers in this study also identified these techniques as the most likely to be used. At the same time, respondents identified the need for more sessions on teaching parent–child interaction to go beyond the theoretical principles and increase caregivers' skills in initiating and maintaining interactive play.

As a central aspect of their learning and engagement, participants emphasised the importance of practitioner modelling, the opportunity for in-session practice and feedback. This corresponds with previous studies that also identified the role of modelling as a core programme component (Dababnah & Parish, 2016; Lachman et al., 2016). Despite this, caregivers asked for additional coaching, indicating a general lack of confidence in using all the taught techniques, which might be one of the drawbacks of delivering many topics within a limited timeframe. Participants also highlighted the value of combining group experience with individualised support during and after the programme. Having an autistic child was experienced by participants as leading to social marginalisation, and programmes of this kind were perceived to provide an inclusive environment for caregivers to normalise their experience and expand their social support networks (Mytton et al., 2014). Practitioners further identified caregiver group support as facilitating programme delivery, as it was perceived to enhance caregiver learning through the sharing of resources and problem-solving strategies. The importance of individualised support was evident in themes about programme flexibility and ongoing communication sustained beyond the duration of programme. Previous qualitative systematic reviews identified a consensus across providers, researchers and parents that such programmes should be tailored to account for individual ability and cultural differences (Durlak & DuPre, 2008; Mytton et al., 2014). However, this presents a challenge to programme delivery, especially given the lack of resources and the absence of research on the key intervention components in LMICs.

Past research has indicated that a collaborative and therapeutic relationship established by knowledgeable and nonjudgmental group leaders contributes to programme success (Mytton et al., 2014). Caregivers in this study also highlighted the dedication of practitioners, who were perceived to be conversant, responsive and unprejudiced, as playing a critical part in their overall positive experience of the programme. Furthermore, multiple caregivers in this programme described how practitioners improved their mental health, suggesting that, in low-resource settings, practitioners do not only teach skills, but provide the type of support that can lead to wider changes in parental functioning.

Having children participating in the sessions was, however, experienced in a number of ways. Although it was identified as a core component in a review on general parent training programmes (Kaminski et al., 2008), it may be more challenging to implement effectively when delivered to parents of autistic children. While children being present in sessions offered substantial opportunities in terms of in vivo practice and feedback, it was also regarded as one of the major barriers to engagement and delivery. Therefore, caregivers commonly requested more playful and structured activities for children when they were not involved in modelling and practice. Childcare needs in parent training programmes for autism may also differ from programmes designed for the general population. Unlike caregivers of children without autism who need regular childcare services to promote attendance (Mytton et al., 2014), due to the core deficits of autism, caregivers of autistic children may expect more structured activities arranged in childcare services to promote child development. For example, parents in a study conducted to evaluate a parent training programme for autistic children in the United States also requested such activities to be included in the programme to engage children in social interactions (Dababnah & Parish, 2016). Moreover, findings from a systematic review of parent training programmes delivered in China to parents of children with a wider range of developmental disabilities, found that components directly targeting child social skills have the potential to further reduce child externalising and international problems (Fang et al., 2020).

With regard to facilitators of participant involvement, findings suggest that caregivers predominantly considered child progress as their major motivation to participate and learn. This is consistent with the quantitative assessment of the SREIA programme process which found a connection between higher levels of engagement and improved child behavioural adjustment (under review). Similarly, a qualitative study in South Africa found that perceived improvement in child development contributed to participant buy-in and might increase the likelihood of behavioural changes in caregivers of autistic children (Makombe et al., 2019).

Qualitative systematic reviews of parent training programmes for children without disabilities have indicated that familial support can influence participant involvement (Butler et al., 2020; Mytton et al., 2014). Caregivers in this study also highlighted the impact of support from other family members on the level of engagement, describing that the lack of support as inhibiting them from signing up for the programme, engaging in the targeted skills practice and maintaining consistent parenting practices. This may also explain the recurrent request for brief training sessions for other family members, alongside the need to enable other family members to be aware of, and to use, the new skills being taught.

Raising an autistic child is stressful irrespective of geographical location, but the severe lack of information and resources (McCabe, 2007) and the long-held negative attitude towards disabilities in China (Huang & Zhou, 2016) may have created additional sources of pressure for the caregivers in the current study. Despite research suggesting that parental mental health can influence treatment adherence (Hackworth et al., 2018), and the fact that the quantitative SREIA assessment found an improvement in mental wellbeing following the course (under review), only a few caregivers explicitly linked their improved mental health status with the course. One potential reason is that many caregivers of children with developmental and behavioural challenges focus less on their own needs, as was shown in the study of Incredible Years for autistic pre-schoolers, where parents reported that there was little time to attend to their own self-care needs (Dababnah & Parish, 2016). However, receiving more formal mental health support was among the most important post-programme needs for caregivers in this study, which may be related to their concerns about the potential fade-out of programme effects in both child and parental domains (van Aar et al., 2017).

Disagreement about the skills being taught was perceived as another barrier to participant involvement. The finding is in line with previous research showing the negative impact of a mismatch between programme philosophy and parental expectations (Lachman et al., 2016), pointing to the importance of clarifying the programme goals at an early stage of intervention. Online delivery was considered by caregivers as less helpful, because of technological failures and the difficulty in maintaining focus. Nonetheless, online delivery of evidence-based intervention has become a global trend driven by the efforts to promoting equal access to healthcare services and accelerated by the COVID-19 pandemic. This highlights the need to address technical challenges, especially in LMICs where there is a lack of infrastructure, and to explore ways to promote participant engagement in digital intervention initiatives.

In terms of programme implementation, research shows that practitioner peer support can promote long-term skill development and the sustainability of autism early interventions in other low-resource contexts (Makombe et al., 2019). Practitioners in this programme recognised the benefits of seeking advice and wider support from peers. However, the lack of organisational resources may have led to their overdependence on peer support. Capacity building has been consistently shown by the literature as a key ingredient to implementation success, especially in terms of the delivery of complex interventions (Mytton et al., 2014). Therefore, it is of particular importance that implementing organisations provide adequate and ongoing supervision and training, especially given that the use of less qualified programme providers has been increasingly advocated to close service gaps in LMIC contexts (Tomlinson et al., 2018). Other aspects of organisational support related to the need for a reduction in their workload and the improvement of practitioner mental health. Despite their dedication, practitioners in this study perceived the quality of their work as being limited by an intensive workload and stress. This was further reflected by their high fidelity score but medium quality of delivery. The stress may be alleviated by, for instance, organisational mental health support and rest breaks. Alternatively, the intensity of the programme can be reduced by spreading out the programme sessions, providing the same number of sessions and content over a slightly longer period of time. Lower intensity may also increase the accessibility of the programme, as well as provide caregivers with longer support and more opportunities to establish and generalise new skills. More fundamental solutions to reduce workload are also needed at a state-level, including investment in autism-related professional training and incentivisation of service provision.

# Strengths and limitations

This study was conducted within a routine service setting and involved participants who were representative of a real-world population, thereby improving the generalisability of the findings. The study nevertheless has several limitations. There was a lack of male caregiver interviewees thereby limiting our ability to comment on their experiences and perceptions. Also, the data were collected retrospectively, which might have introduced recall bias. In addition, the interviews were conducted in Chinese thereby necessitating translation, which could have resulted in more bias. To improve validity, the data analysis was conducted in Chinese, involving two bilingual coauthors, who were born and raised in China. The English report was prepared by one bilingual co-author in discussion with the other, and further checked by co-authors who are native English speakers. Also, we did not identify distinct patterns between caregivers with different rates of attendance and engagement, potentially due to the high levels of involvement in general. Future studies may consider assessing participant involvement using standardised tools to increase sensitivity. Future research can also stratify participants by demographic characteristics and explore how participant experience may differ by background. Moreover, the involvement of stakeholders in this study was insufficient. Stakeholders were limited to organisation staff, who provided only basic inputs into the research design and interpretation of results. Family beneficiaries and stakeholders in the wider community could have been included. More opportunities could also have been sought at every stage to deepen the partnership, so that we could further improve the relevance of the research to the community and shorten the time it took to translate the findings into practice.

# Conclusion

This study provides important insights on key stakeholder perceptions about a short-term intensive parent training programme for parents of autistic children in an LMIC. A number of suggestions are provided for future parent training programmes for families of autistic children. In terms of programme design, adequate opportunities for caregivers to practice and receive feedback, as well as a mixture of group and individualised support, are central to creating an overall positive and empowering experience. It is also recommended that support be extended beyond the duration of the programme, or the programme sessions be spread out over a slightly longer period of time, to maintain programme effects. Autism knowledge and programme goals should be introduced at an early stage to help caregivers maintain focus and increase the likelihood of collaborative relationships between practitioners and caregivers. More instructions on parent-child interaction may also promote parent-child relationships and parental use of targeted skills at home. Structured activities can be arranged separately for children, except when parents need to practice skills directly with children with the support of a practitioner. Resources should be shared to help caregivers navigate the service system and improve mental health. Other family members may also need to be briefly involved to strengthen family support. In addition, helping caregivers identify progress in children may promote participant engagement. To improve the quality of programme delivery, more organisational resources including adequate and ongoing supervision and training, and mental health support - should be provided to practitioners. Practitioner stress could also be reduced by spreading out the programme sessions. Support at a national level is needed to address the human resource challenges. Finally, creative solutions are needed to overcome technical barriers and to engage participants via online platforms. More mixedmethods research is needed to evaluate the impact of the recommended changes and assess their effectiveness using experimental designs, such as micro-trials and factorial experiments.

#### **Declaration of conflicting interests**

The author(s) declared no potential conflicts of interest with respect to the research, authorship and/or publication of this article.

# Funding

The author(s) disclosed receipt of the following financial support for the research, authorship and/or publication of this article: ZF's involvement was partly funded by the Swire Charitable Trust. JML's involvement was supported by the University of Glasgow Social and Public Health Sciences Unit Complexity and Relationships in Health Improvement Programs of the Medical Research Council MRC UK and Chief Scientist Office (Grant: MC\_UU\_00022/1 and CSO SPHSU16, MC\_UU\_00022/3 and CSO SPHSU18). The funders played no role in the design, conduct, or interpretation of the analyses.

# ORCID iD

Zuyi Fang (D) https://orcid.org/0000-0002-8619-0137

#### Supplemental material

Supplemental material for this article is available online.

#### References

- Baydar, N., Reid, M. J., & Webster-Stratton, C. (2003). The role of mental health factors and program engagement in the effectiveness of a preventive parenting program for Head Start mothers. *Child Development*, 74(5), 1433–1453. https://doi.org/10.1111/1467-8624.00616
- Berkel, C., Mauricio, A. M., Schoenfelder, E., & Sandler, I. N. (2011). Putting the pieces together: An integrated model of program implementation. *Prevention Science*, *12*(1), 23–33. https://doi.org/10.1007/s11121-010-0186-1
- Boyatzis, R. E. (1998). *Transforming qualitative information: Thematic analysis and code development*. SAGE.
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77–101. https://doi.org/10.1191/1478088706qp063oa
- Butler, J., Gregg, L., Calam, R., & Wittkowski, A. (2020). Parents' perceptions and experiences of parenting Programmes: A systematic review and metasynthesis of the qualitative literature. *Clinical Child and Family Psychology Review*, 23(2), 176–204. https://doi.org/10.1007/s10567-019-00307-y
- Carr, T., Shih, W., Lawton, K., Lord, C., King, B., & Kasari, C. (2016). The relationship between treatment attendance, adherence, and outcome in a caregiver-mediated intervention for low-resourced families of young children with autism spectrum disorder. *Autism*, 20(6), 643–652. https:// doi.org/10.1177/1362361315598634
- Clark, E., Zhou, Z., & Du, L. (2019). Autism in China: Progress and challenges in addressing the needs of children and families. *International Journal of School and Educational Psychology*, 7(2), 135–146. https://doi.org/10.1080/216836 03.2019.1570885
- Courchesne, E., Campbell, K., & Solso, S. (2011). Brain growth across the life span in autism: Age-specific changes in anatomical pathology. *Brain Research*, 1380, 138–145. https:// doi.org/10.1016/j.brainres.2010.09.101
- Creswell, J. W., Klassen, A. C., Clark, V. L. P., & Smith, K. C. (2011). Best practices for mixed methods research in the health sciences. National Institutes of Health.
- Dababnah, S., & Parish, S. L. (2016). Feasibility of an empirically based program for parents of preschoolers with autism spectrum disorder. *Autism*, 20(1), 85–95. https://doi. org/10.1177/1362361314568900
- Durlak, J. A., & DuPre, E. P. (2008). Implementation matters: A review of research on the influence of implementation on program outcomes and the factors affecting implementation. *American Journal of Community Psychology*, 41(3–4), 327–350. https://doi.org/10.1007/s10464-008-9165-0
- Fang, Z., Barlow, J., & Zhang, C. (2020). Parenting programs that address physical abuse in childhood for families of children with developmental disabilities in Mainland China: Systematic review and meta-regression. *Trauma*, *Violence & Abuse*. Advance online publication. https://doi. org/10.1177/1524838020915599

- Fereday, J., & Muir-Cochrane, E. (2006). Demonstrating rigor using thematic analysis: A hybrid approach of inductive and deductive coding and theme development. *International Journal of Qualitative Methods*, 5(1), 80–92. https://doi. org/10.1177/160940690600500107
- Glaser, B. G., & Strauss, A. (1967). *The discovery of grounded theory: Strategies for qualitative research*. Aldine.
- Green, J., & Thorogood, N. (2014). *Qualitative methods for health research*. SAGE.
- Guo, D., Deng, X., Zhao, Q., Wen, H., & Huang, J. (2014). The needs of parents of children with autism in China [孤独症家长需求 分析与对策建议]. *Disability Research*, 2, 43–48. https://kns. cnki.net/KCMS/detail/detail.aspx?dbcode=CJFD&dbname=-CJFD2014&filename=CJRY201402011&v=MDg2NDZHN Eg5WE1yWTIFWIISOGVYMUx1eFITN0RoMVQzcVRyV 00xRnJDVVJMT2ZZdWRvRnIIZ1ZyL0pKaWZaZDc=
- Hackworth, N. J., Matthews, J., Westrupp, E. M., Nguyen, C., Phan, T., Scicluna, A., Cann, W., Bethelsen, D., Bennetts, S. K., & Nicholson, J. M. (2018). What influences parental engagement in early intervention? Parent, program and community predictors of enrolment, retention and involvement. *Prevention Science*, 19(7), 880–893. https://doi. org/10.1007/s11121-018-0897-2
- Hayes, S. A., & Watson, S. L. (2013). The impact of parenting stress: A meta-analysis of studies comparing the experience of parenting stress in parents of children with and without autism spectrum disorder. *Journal of Autism and Developmental Disorders*, 43(3), 629–642. https://doi. org/10.1007/s10803-012-1604-y
- Huang, M., & Zhou, Z. (2016). Perceived self-efficacy, cultural values, and coping styles among Chinese families of children with autism. *International Journal of School and Educational Psychology*, 4(2), 61–70. https://doi.org/10.10 80/21683603.2016.1130562
- Kaminski, J. W., Valle, L. A., Filene, J. H., & Boyle, C. L. (2008). A meta-analytic review of components associated with parent training program effectiveness. *Journal of Abnormal Child Psychology*, 36(4), 567–589. https://doi.org/10.1007/ s10802-007-9201-9
- Lachman, J. M., Sherr, L. T., Cluver, L., Ward, C. L., Hutchings, J., & Gardner, F. (2016). Process evaluation of a parenting program for low-income families in South Africa. *Journal* of Child and Family Studies, 25(7), 2337–2352. https://doi. org/10.1007/s10826-016-0389-6
- Makombe, C. B. T., Shabalala, N., Viljoen, M., Seris, N., Vries, P. J., & De Franz, L. (2019). Sustainable implementation of early intervention for autism spectrum disorder through caregiver coaching : South African perspectives on barriers and facilitators. *Pediatric Medicine*, 2, Article 39. https:// doi.org/10.21037/pm.2019.07.08
- McCabe, H. (2007). Parent advocacy in the face of adversity: Autism and families in the People's Republic of China. Focus on Autism and Other Developmental Disabilities, 22(1), 39– 50. https://doi.org/10.1177/10883576070220010501
- McCabe, H. (2012). Bamboo shoots after the rain: Development and challenges of autism intervention in China. *Autism*, 17(5), 510–526. https://doi.org/10.1177/1362361312436849
- Morris, J., Belfer, M., Daniels, A., Flisher, A., Villé, L., Lora, A., & Saxena, S. (2011). Treated prevalence of and mental health services received by children and adolescents in 42 lowand-middle-income countries. *Journal of Child Psychology*

*and Psychiatry and Allied Disciplines*, *52*(12), 1239–1246. https://doi.org/10.1111/j.1469-7610.2011.02409.x

- Mytton, J., Ingram, J., Manns, S., & Thomas, J. (2014). Facilitators and barriers to engagement in parenting programs: A qualitative systematic review. *Health Education and Behavior*, 41(2), 127–137. https://doi.org/10.1177/1090198113485755
- National Bureau of Statistics of China. (2021). The Seventh National Population Census of the People's Republic of China. http:// www.stats.gov.cn/tjsj/pcsj/rkpc/6rp/indexch.htm
- Nevill, R. E., Lecavalier, L., & Stratis, E. A. (2018). Metaanalysis of parent-mediated interventions for young children with autism spectrum disorder. *Autism*, 22(2), 84–98. https://doi.org/10.1177/1362361316677838
- Nix, R. L., Bierman, K. L., & McMahon, R. J., & The Conduct Problems Prevention Research Group. (2009). How attendance and quality of participation affect treatment response to parent management training. *Journal of Consulting and Clinical Psychology*, 77(3), 429–438. https://doi. org/10.1037/a0015028
- Oono, I. P., Honey, E. J., & McConachie, H. (2013). Parentmediated early intervention for young children with autism spectrum disorders (ASD). *Evidence-Based Child Health:* A Cochrane Review Journal, 8(6), 2380–2479. https://doi. org/10.1002/14651858.CD009774.pub2.www.cochranelibrary.com
- Parsons, D., Cordier, R., Vaz, S., & Lee, H. C. (2017). Parentmediated intervention training delivered remotely for children with autism spectrum disorder living outside of urban areas: Systematic review. *Journal of Medical Internet Research*, 19(8), Article e6651. https://doi.org/10.2196/ jmir.6651
- Postorino, V., Sharp, W. G., McCracken, C. E., Bearss, K., Burrell, T. L., Evans, A. N., & Scahill, L. (2017). A systematic review and meta-analysis of parent training for disruptive behavior in children with Autism Spectrum Disorder. *Clinical Child and Family Psychology Review*, 20(4), 391– 402. https://doi.org/10.1007/s10567-017-0237-2
- Ratliff-Black, M., & Therrien, W. (2020). Parent-mediated interventions for school-age children With ASD: A meta-analysis. Focus on Autism and Other Developmental Disabilities, 36(1), 3–13. https://doi.org/10.1177/1088357620956904
- Strauss, K., Mancini, F., & Fava, L. (2013). Parent inclusion in early intensive behavior interventions for young children with ASD: A synthesis of meta-analyses from 2009 to 2011. *Research in Developmental Disabilities*, 34(9), 2967–2985. https://doi.org/10.1016/J.RIDD.2013.06.007
- Sun, X., Allison, C., Wei, L., Matthews, F. E., Auyeung, B., Wu, Y. Y., Griffiths, S., Zhang, J., Baron-Cohen, S., & Brayne, C. (2019). Autism prevalence in China is comparable to Western prevalence. *Molecular Autism*, 10(1), Article 7. https://doi.org/10.1186/s13229-018-0246-0
- Tomlinson, M., Hunt, X., & Rotheram-Borus, M. J. (2018). Diffusing and scaling evidence-based interventions: Eight lessons for early child development from the implementation of perinatal home visiting in South Africa. Annals of the New York Academy of Sciences, 1419(1), 218–229. https:// doi.org/10.1111/nyas.13650
- Totsika, V., Hastings, R. P., Emerson, E., Lancaster, G. A., & Berridge, D. M. (2011). A population-based investigation of behavioural and emotional problems and maternal mental health: Associations with autism spectrum disorder and

intellectual disability. *Journal of Child Psychology and Psychiatry and Allied Disciplines*, 52(1), 91–99. https://doi. org/10.1111/j.1469-7610.2010.02295.x

- van Aar, J., Leijten, P., de Castro, O. B., & Overbeek, G. (2017). Sustained, fade-out or sleeper effects? A systematic review and meta-analysis of parenting interventions for disruptive child behavior. *Clinical Psychology Review*, 51, 153–163. https://doi.org/10.1016/j.cpr.2016.11.006
- Walker, S. P., Wachs, T. D., Grantham-McGregor, S., Black, M. M., Nelson, C. A., Huffman, S. L., Baker-Henningham, H., Chang, S. M., Hamadani, J. D., Lozoff, B., Gardner, J. M. M., Powell, C. A., Rahman, A., & Richter, L. (2011). Inequality in early childhood: Risk and protective factors for early child development. *Lancet*, 378(9799), 1325– 1338. https://doi.org/10.1016/S0140-6736(11)60555-2
- Warren, Z., McPheeters, M. L., Sathe, N., Foss-Feig, J. H., Glasser, A., & Veenstra-VanderWeele, J. (2011). A systematic

review of early intensive intervention for autism spectrum disorders. *Pediatrics*, *127*(5), e1303–e1311. https://doi .org/10.1542/peds.2011-0426

- Whittingham, K., Sofronoff, K., & Sheffield, J. K. (2006). Stepping Stones Triple P: A pilot study to evaluate acceptability of the program by parents of a child diagnosed with an Autism Spectrum Disorder. *Research in Developmental Disabilities*, 27(4), 364–380. https://doi.org/10.1016/j. ridd.2005.05.003
- Zwaigenbaum, L., Bauman, M. L., Choueiri, R., Kasari, C., Carter, A., Granpeesheh, D., Mailloux, Z., Roley, S. S., Wagner, S., Fein, D., Pierce, K., Buie, T., Davis, P. A., Newschaffer, C., Robins, D., Wetherby, A., Stone, W. L., Yirmiya, N., Estes, A., . . . Natowicz, M. R. (2015). Early intervention for children with autism spectrum disorder under 3 years of age: Recommendations for practice and research. *Pediatrics*, *136*, S60–S81. https://doi.org/10.1542/peds.2014-3667E